



State of the Art in Designing Fish-Friendly Turbines: Concepts and Performance Indicators

Phoevos (Foivos) Koukouvinis 💿 and John Anagnostopoulos *

Laboratory of Hydraulic Turbomachines, Department of Mechanical Engineering, National Technical University of Athens, Heroon Polytechniou 9, 157 80 Zografou, Greece

* Correspondence: anagno@fluid.mech.ntua.gr; Tel.: +30-210-7721080

Abstract: The expanding role of renewable energy sources in the electricity market share implies the increasing role of hydropower and the exploitation of unharnessed hydraulic potential, in the scope of sustainability and net zero emissions. Hydro-turbine design practices are expected to expand beyond achieving high efficiency goals, to multi-objective criteria ranging from efficient reversible operation to fish-friendly concepts. The present review paper outlines fundamental characteristics of hydropower, summarizing its potential impact toward aquatic life. Estimates of lethality for each damage mechanism are discussed, such as barotrauma, blunt impact and shearing, along with relevant advances in experimental techniques. Furthermore, numerical techniques are discussed, ranging from simple particle tracking to fully coupled six-degree-of-freedom tracking, which can be used to investigate candidate designs and their fish-friendly performance, presenting their advantages and disadvantages. Subsequently, a link to the individual damage mechanisms is established, to proposed holistic performance metrics, useful for providing estimates of fish-friendliness of a given hydropower installation. Finally, recent developments and design practices for fish-friendly turbine concepts are presented.

Keywords: fish-friendly hydropower; hydropower statistics; fish injury assessment; experimental methods; numerical methods; turbine design considerations



Citation: Koukouvinis, P.; Anagnostopoulos, J. State of the Art in Designing Fish-Friendly Turbines: Concepts and Performance Indicators. *Energies* **2023**, *16*, 2661. https://doi.org/10.3390/en16062661

Academic Editor: Andrzej Teodorczyk

Received: 1 February 2023 Revised: 20 February 2023 Accepted: 7 March 2023 Published: 12 March 2023



Copyright: © 2023 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/).

1. Introduction

Hydropower is perhaps the oldest form of renewable energy resource, and it plays a paramount role in energy production worldwide. Indicatively, Figure 1 shows the contribution of hydropower, compared to other resources globally; in terms of global energy consumption (that is, including transportation), hydropower contributes 7% of the total, whereas in terms of electricity production, hydropower contributes almost 15% of the total (see also [1]). It is also notable that the contribution is effectively comparable to all the other renewable energy resources combined; indicatively, it has an estimated contribution of almost 60% compared to other renewable energies based on 2021 data [2].

The role of hydropower has been constantly increasing over the years. Indicatively, as shown in Figure 2, the recent trend both in terms of installed capacity and produced energy per year shows an annual growth of 1.8% and 2.3%, on average for the last 2–3 years, respectively. Despite the minor drop in annual growth of installed capacity for the year 2019 and, given the goal for a Net Zero Emissions by 2050, hydropower is expected to maintain an average annual generation growth rate of about 3% in 2022–2030 [3]. The growing role of hydropower is justified for two main reasons:



Figure 1. (a) Global energy consumption by fuel source and (b) global electricity generation by fuel type (source, BP Energy Statistics 2022 [1]).







Figure 2. Hydropower statistics and trends over the recent years: (a) Installed capacity and (b) annually produced energy. (source, IRENA Renewable Energy Statistics 2022 [2]).

(A) It can greatly contribute as an energy storage solution, in the form of pumped hydropower (also abbreviated as PHP), along with batteries, allowing for the increased penetration of other renewable sources [4], such as wind and solar energies. Pumped hydropower not only generates electricity and serves as daily or weekly balancing, but also provides additional services to the grid, such as system inertia, frequency response and grid regulation. Indicatively, as of 2020 statistical data [5], pumped hydropower contributes more than 90% in installed capacity. Estimates indicate that the installed capacity of reversible hydro/pumped storage will increase by 42% until 2026 [6].

(B) There is a considerable quantity of unharnessed hydropower potential, with rough estimates indicating that about 25% of the technically achievable and 50% of the economically feasible potential (for the current technologies) is presently being utilized, see [7–9]. Indicatively, an appreciable hydraulic potential can also be found in existing infrastructure that currently lacks generating units (e.g., existing barrages, weirs, dams, canal fall structures, water supply schemes) by adding new hydropower facilities.

Despite the renewable nature of hydropower, it unavoidably has an impact on the local ecosystems and aquatic life. In particular, the following aspects have been identified and recognized [10]:

- The direct destruction of fauna and flora habitats, by changing the use of land and by submerging large areas, to achieve the formation of a reservoir;
- Indirect disturbance to local life, due to building infrastructure (e.g., roads, electrical grid, etc.) and gaining access to previously inaccessible areas. This can also involve induced disturbance due to population living in the reservoir area being resettled to formerly natural areas;
- Disruption of the natural flow of a water stream, by forming a reservoir. Inherently, the construction of any sort of dam implies a partial or total impedance of free migration of aquatic species, the blocking of sediments that would replenish downstream ecosystems and the change in the downstream flow patterns of the river.

The first two points mainly affect terrestrial vegetation and fauna, while the third point refers to impacts on hydrobiology (fish and other aquatic life). Whereas there are methodologies for assessing the impact both on terrestrial vegetation, as well as fauna, and taking proper mitigation measures, see indicatively [10], fish fauna is considered to be particularly susceptible to the disruption caused by hydropower facilities [11,12]. In particular, the presence of the dam and the introduction of the hydraulic turbine, necessary to extract energy from the water stream, inherently alter the natural course of a river stream and affect the active (migration) and passive (drift) movements of aquatic organisms' travelling routes. In the last decade, severe declines in the freshwater populations has been observed and a major contribution to this decline is believed to be due to river ecosystem fragmentation [13,14]. Undoubtedly, the fish fauna of a river is, in every case, affected in a major way by a hydropower project; hence, it has to be one major focal point of design considerations.

Over time, measures for enabling fish passage through a hydropower project have been developed in the form of fish passages, a term generally used to refer to any structure built to facilitate the travel of fish through a water stream. These can be classified as:

(A) Passive, in the form of:

- Artificial fishways incorporate artificial flow reduction elements such as baffles (also known as Denil fishways) or steps (e.g., pool-and-weir, vertical-slot fishways or fishladders, see Figure 3a,b, respectively);
- Nature-like fishways contain natural features that increase bottom roughness such as cobble and boulders (see, e.g., Figure 4a), although they may incorporate some engineered elements such as anchored concrete blocks or other artificial elements that may be found in technical fishways.



(a)



(b)





(a)

(b)



(B) Active systems, such as fish locks and fish lifts that use mechanical locking gates to direct fish and lifting devices such as baskets to physically move them past barriers, see, e.g., Figure 4b.

Nevertheless, despite these measures which enable fish to pass through the installed infrastructure, systematic studies have demonstrated that these are not perfect remedies [15]; in fact, extensive statistical analysis of fish mortality indicates that bypasses pose a decreased injury risk relative to controls, whereas turbines and spillways were associated with the highest injury risks relative to controls [16]. Furthermore, despite the existence of general guidelines or risk factors for aquatic life (e.g., Kaplan turbines are known to be safer comparing to Francis turbines [16]), the development of methodologies estimating damage/mortality [17–19] or the compilation of extensive statistical data on mitigation measures and fish mortality (see [15,20–25]), there is lack of a widely accepted indicator or methodology that could characterize reliable fish-friendliness. Such an assessment is critical in determining the effectiveness of potential remedies to alleviate risks, since fish passage is highly dependent on details of each particular site and fish species [16].

As underlined at the beginning of this section, it is clear that hydropower will continue playing an important role in power generation either in the form of production or energy storage and grid regulation. Moreover, it is expected that, in the future, previously unharnessed water resources (such as unpowered dams, unexploited low-head streams, small reservoirs, streams of impaired quality, etc., all the previous also denoted as "hidden hydro"), will be further exploited (see for example [26–28], or the recent HORIZON-CL5-2021-D3-03-11 call [29]). Hence, hydropower penetration in the near future is expected to place more pressure on aquatic life [30,31], rendering fish-friendly design concepts essential considerations for any prospective hydropower infrastructure (see [32,33]). The aforementioned observation is the motivation behind the present paper, aiming to condense existing know-how on fish-friendly hydropower concepts, identify challenges and propose further fields of study.

The present paper is structured as follows: Section 2 provides a brief overview of the structure of a hydropower facility, Section 3 presents potential mechanisms and phenomena due to hydraulic machinery operation that can adversely affect fish, along with relevant biological investigations on the topic and Section 4 examines methodologies for predicting fish injury, derived through experiments or simulations. Then, Section 5 discusses attempts on deriving fish-friendliness indices and Section 6 investigates fish-friendly turbine concepts. Finally, Section 7 is the conclusion, discussing challenges and future perspectives.

The methodology for performing the literature review involved a literature search to six electronic journal databases relevant to engineering, energy and environmental sciences, as well as the US Department of Energy resources and Google Scholar. The search was performed using terms such as fish-friendly hydropower, environmentally friendly turbines, fish barotrauma, turbine blade strike, fish injury mechanism and turbine design optimization. The literature resources were included if relevant to the scope of each section of the present paper.

2. Structure of Hydropower Station

For the sake of completeness, a short description of a typical hydropower plant is depicted in Figure 5 and is discussed to demonstrate its main components and mitigation measures for aquatic life; it will also assist in the understanding of damage mechanisms, which will be discussed in Section 3.

Inherently to their method of operation, hydroelectric power plants direct a water stream through the hydraulic turbine to extract energy [34]. This can be achieved in two ways: in the form of impoundment hydroelectric power plants or in the form of diversion hydroelectric power plants.

The first type, shown in Figure 5a, implies the construction of an obstruction, termed as dam. This obstruction cuts off the normal water path, forming a reservoir and forcing the main water stream to pass through the turbine. This type of hydropower implies heavy capital costs due to large civil engineering projects and is commonly associated with large hydropower facilities (>15 MW [34]). Such facilities also include reversible hydropower plants in the form of pumped storage systems, if the turbine is designed to be operated as a pump. Apart from the infrastructure for directing water through the turbine, common additional components involve regulating systems, such as control gates to regulate incoming flow, or safety systems, such as spillways in case hydraulic energy needs to be rejected. Furthermore, to protect aquatic life, it is common to introduce grates or racks at the intake pipe of the turbine to prevent large objects/fish to enter the turbine, potentially causing damage; recently, electric [35,36], electromechanical [37] or acoustic [38] deterrence techniques are also proposed to prevent fish from entering the turbine intake. Additionally, fishways/fish ladders may be incorporated at a bypass stream.

In the diversion design the main path of the water stream remains unaffected, but only a small, diverted flow actually passes through the hydraulic turbine and then merges further downstream with the main river, see Figure 5b. Additional preventive measures, such as intake grates, may be employed to prevent large fish entering the turbine intake as well. Naturally, this type of project is more fish-friendly, since the majority of the water stream remains relatively unaffected; however, a large part of the water stream remains inherently unexploited. Hence, it is suitable for small hydropower projects (<15 MW [34]). Furthermore, pumped storage is not an option for this type of project, since there is no upper reservoir to store water.

In terms of turbine technologies, it is common to categorize hydropower facilities depending on the available head, commonly indicated with the symbol *H*. Commonly, low-head plants use bulb turbines (H < 20 m) or Kaplan (20 m < H < 60 m). Mediumto-high head plants (40 m < H < 400 m) use Francis turbines (see Figure 5c) and finally plants of very high head (350 m < H < 1100 m) use Pelton turbines (see Figure 5d). Many variations or alternatives to the aforementioned turbine types exist, such as Turgo, which is similar to Pelton, or the Archimedes screw design which can be an alternative to bulb



turbines for very low heads (1 m < *H*); some of them will be discussed in Section 7 as more fish-friendly alternatives.

Figure 5. Typical structure of hydroelectric power plant with the main components shown (**a**) impoundment type (**b**) diversion type (Credit: U.S. Department of Energy). Images adapted from [39]. Typical turbines in hydropower: (**c**) Francis turbine (adapted from [40]) (**d**) an indicative view of a Pelton impulse turbine.

A distinct classification of the turbine designs, however, can be made in terms of reaction and impulse turbines. The distinction lies at the location of the pressure drop, whether it takes place at the stator (impulse turbine) or the rotor (reaction turbines), as this affects the operation of the turbine. Impulse turbines, such as Pelton turbines, operate inherently in a non-immersed environment. Hence, they cannot play the role of a reversible turbine, unless a dedicated pump is also used [41].

3. Mechanisms of Fish Damage and Mortality during Passage through a Turbine

In order to define criteria for the design of fish-friendly turbines, it is necessary to identify the mechanisms relevant to fish damage. Such mechanisms have been identified since the early 2000s, in an effort to improve designs, provide guidelines and develop novel turbine concepts with minimum effect on passing fish. In particular, several critical damage mechanisms have been identified [24,42–44], which are presented below, following the flow direction from the inlet racks toward the draft tube (see also Figure 6).



Figure 6. Illustration of locations where fish damage may occur (adapted from [45]).

3.1. Pressure Changes

As a passing fish moves through the turbine intake, through the turbine, the draft tube and finally downstream the tailrace, it experiences severe pressure changes. After the racks and inside the penstock, pressure gradually increases, due to hydrostatic pressure [46]; an indicative of such variation is shown in Figure 7a; however, depending on the elevation change, this can be in the order of 1 bar to more than 10 bar. Then, at the vicinity of the stator blades, and especially the runner, pressure drops rapidly, much faster than the pressure rise. Gauge pressure can drop even below zero (vacuum), near the suction side of the blades [47,48], or the spiraling vortex tube, also known as vortex rope in the relevant literature, emanating from the turbine hub [40,49]; indeed, the latter is well known to induce strong low-frequency vibrations to the whole turbine, which potentially threaten the integrity of the whole installation [50–53]. In any case, as the trajectory of a passing fish reaches the tailrace, pressure recovers gradually to near atmospheric conditions.



Figure 7. Cont.





Figure 7. (a) An indicative pressure variation (orange line, based on data from [46]) that a fish experiences as it passes through a Kaplan turbine (fish track is the dashed line). (b) An indicative pressure variation and (c) shear stress magnitude, as a fish passes through an axial turbine (based on authors' work, see Section 4.2.3). In (b), the dashed–dotted (sloped) line indicates the depressurization rate of 90 kPa/s where the onset of mortality is observed [54]. The dashed–dotted (horizontal) line at the bottom indicates the onset of cavitation. In (c), the dashed–dotted (horizontal) line indicates a threshold of 35 Pa, which was found damaging for fish larvae [55]. (d) The indicative trajectory is examined and colored according to time. Black indicates the stator vanes, red the rotor blades.

Given the fact that fish are equipped with a gas bladder to adjust buoyancy, it is expected that pressure changes will greatly affect their capability to float and adjust their motion. Nevertheless, extensive research with controlled pressurization and depressurization of fish [56–59], both in terms of magnitude and rate, has shown that, in general, a pressure increase is rather well received; even at high pressurization rates, fish damage is minimal and no direct mortality was observed [43]. On the other hand, depressurization, and especially at a fast rate (indicative values for onset of mortality in the order of 90 kPa/s [54]), is rather problematic, leading to barotrauma [60,61]. Typical effects of barotrauma-related injuries include ruptured swim bladder, exopthalmia, internal hemorrhaging and gas bubbles (emboli). Extensive studies, using X-rays at fish exposed to rapid decompressions [60], indicated that the most prevalent theory for fish barotrauma is the rupturing of the fish gas bladder, which subsequently releases gas inside the bloodstream causing the aforementioned effects, eventually leading to severe mortality. While factors such as the pressure history (e.g., acclimation at high pressures) or the structure or existence of the fish gas bladder (e.g., physostomous fish can expel gas from the bladder naturally; lampreys lack gas bladder; both cases are expected to be more resilient to decompression [60]), an extensive compilation of experimental studies indicated that decompression mortalities can even reach 100% [43].

In cases of local pressure drops at vapor pressure levels, a more severe case of barotrauma may occur due to cavitation. Given the highly transient nature of cavitation and the associated rapid decompression and pressure recovery, formed vaporous pockets tend to collapse violently; indicatively, cavitation collapse is known to produce very fast microjets [62] that can pierce tissues [63] and even pit metallic surfaces [64] and to emit powerful shockwaves [65]. Despite the resulting aggressiveness of cavitation collapse, experimental investigations have shown mixed results [66], with a range of mortalities between 0 and 60% when fish are exposed to cavitating conditions. Nevertheless, there is a limited number of investigations on fish passage through a turbine in association with cavitation effects; possibly, this is because turbines are designed or controlled to limit operation in cavitating conditions [43], as this has severe adverse effects on their operation, longevity and efficiency.

3.2. Mechanical Damage

A more direct cause of fish mortalities is the interaction of fish with the rigid, structural/operational elements of the turbine rotor/stator. This can be classified into three categories, as: (1) strike, in the case of fish impact on surfaces, (2) grinding, in the case of fish being trapped in small gaps or clearances and (3) friction and abrasion.

Mechanical injury due to strike is expected to depend on various parameters, such as the fish size, the turbine passage, blade shape, rotating speed and flow rate. Indeed, numerous studies ([22,67–69]) have attempted the derivation of correlations for estimating the effect of the aforementioned factors on fish mortality; however, simplifying assumptions are commonly being involved, and thus confidence varies greatly [43]. Moreover, such estimations may be skewed by the exact fish species' behavior and characteristics. For example, fish behavior affects the outcome of the impact, since some fish species tend to move as a rigid body through the turbine, whereas other tend to react and actively avoid collision [43]. Additionally, aside the influence of fish dimensions, fish weight can play a role in the impact outcome, since smaller fish have less inertia; hence, they tend to be dragged by the flow around the blades, instead of colliding with them. Indicatively, studies indicated that fish with a mass of less than 20 g were unlikely to suffer a blade collision [70]. Nevertheless, extensive testing with different blade profiles and rotational velocities has shown that blunt, thick blades rotating at low velocities are the least threatening to fish [70,71]. Indicatively, collision velocities at ~5 m/s with thick blades were found to cause little damage and no mortality, whereas impacts on thinner blades at collision velocities of 6–7 m/s caused severe damage.

Considering grinding, there are very few references or investigations relevant to its influence. While it is directly related to the existence of gaps and clearances, where passing fish may get entrained, it is considered to be related more to fish dragged by leakage flows (e.g., at various control gates), rather than the main flow path. Hence, a general guideline involves eliminating gaps, as minimizing them at a size of less than 2 mm is expected to prevent fish injuries due to grinding [72].

The last category of mechanical damage, friction and abrasion refers to the shearing of fish along the solid turbine walls. As an effect, it is observed mainly in the draft tube [43] and is related to the swirling motion, imparted to the passing flow due to turbine operation. Despite the existence of dedicated techniques to investigate the occurrence and extension of external damage [73], it is, in practice, rather difficult to distinguish friction/abrasion from the other types of physical damage [74]. Efforts to devise techniques for the quantification and distinction of abrasion from shearing are ongoing [75]. In any case, the existence of irregularities on the draft tube, such as protrusions from welding, or bolts used for pipe connections, greatly increases the potential risk of abrasion [74].

3.3. Shear Stresses

The flowing water streams inherently develop from boundary layers, recirculations and flow detachments, all attributed to viscosity effects; thus, shear stresses are induced in all these cases due to the interaction of fluid layers moving at different velocities. Indicatively, shear stresses in water streams/rivers may be in the order of 100 Pa, whereas in the draft tubes of turbines they are in the order of 5000 Pa [76]. Even though these values are much lower than pressure magnitude (which can be two or three orders larger—see indicatively Figure 7b,c), fish are much more sensitive to shear, since exposure to high shear levels can be comparable to exposure to friction forces generated by two solid surfaces contacting and sliding against each other [77]. Common injuries resulting from high levels of shear are loss of mucous layer, descaling, tissue tearing or bruising and even decapitation [78].

Quantification of shear thresholds for fish damage has been performed in the past with various methods. The most common approach involves a high-velocity jet that is expelled in a stagnant water container, in which test subjects are kept. The jet produces a shear layer around its axis and shear stresses can be quantified in terms of the radial distance from the center of the jet and the axial distance from the source of the jet. This technique has been employed several times to identify damage/mortality thresholds (see indicatively [70,79,80]), with values exceeding 3400 Pa being tested; damage was strongly dependent on fish species (indicatively, eels seemed to be the most resilient), though a threshold of 1600 Pa was proposed for onset of descaling [81]. A drawback of this method is that it does not allow for a precise control of the actual shear stress fish are exposed to, since shear stress magnitude is strongly related to exact position around the jet axis. Additionally, the formation of a fast-moving jet may involve the generation of shear layer cavitation [43,80], which can further blur the actual influence of shear stresses as a damage mechanism. Alternative approaches involve a Taylor–Couette-type apparatus [55], where concentric cylinders form an annular passage filled with water and test subjects, though it has been tested on fish larvae, demonstrating mortalities at shear magnitudes of even 35 Pa.

3.4. Vorticity and Turbulence

The flow in hydraulic turbines and relevant piping can be classified as turbulent, given that the Reynolds number is in the order of 10^6 – 10^7 , or even more. Turbulent flows inherently imply a highly fluctuating unsteady flow, with many different scales of motion; in practice, this means that there are strong variations of local pressure and velocity which can cause strain to the body of passing fish [43], either in the form of normal or shear stresses, respectively. Hence, high turbulence levels are expected to cause mortalities for both large fish and their eggs or larvae [82].

Despite the existence of limited investigations on correlating turbulence and fish mortality, its influence is not clearly understood. There are very few investigations relating turbulence directly with fish mortality; indicatively, in [83] the authors employed a water jet to create a turbulent flow field of variable intensity, in an annular passage, in which fish larvae and eggs were exposed for different durations. A near to 80% mortality rate was found, after a 10 h exposure to high turbulence conditions, which were quantified as velocity fluctuations of ~0.55 m/s and pressure fluctuations of ~6300 dynes/cm² (~630 Pa). The turbulence level was found to contribute more to lethality than to frequency of disturbance.

In other works, turbulence is considered as only indirectly affecting survivability, given that it contributes to the disorientation of passing fish at the tailrace [15,84]; these fish are subsequently more susceptible to being preyed upon. Indeed, turbulence and shear stress influence is considered to be the most difficult to describe [77].

4. Assessing Fish Damage

4.1. Experimental Methods

Numerous studies have been conducted with experimental means to understand the mechanisms of fish passage and potential damage and to extrapolate these findings to fish-friendly turbine concepts. These mainly involved tagging fish with various means, followed by their subsequent release upstream a specific hydropower plant, while also tracking their position and their recapture further downstream.

An example of such investigations involved the use of inflatable balloons, to ensure the recapture of fish passing through the turbine. In particular, the inflatable tagging device was externally mounted and triggered before release. A delayed chemical reaction, whose timing can be pre-adjusted accordingly, released gas inflating the balloon tag shortly after turbine passage, facilitating easy tracking and recapture for further examination. This technique allowed for up to 96% of tagged fish being recaptured and the derivation of statistics for Kaplan and mixed-flow turbine types, showing a low percentage of injury. In particular, one study showed that only 5% of the tested fish exhibiting small lacerations [85], whereas another study found that only 5.7% of the tested fish had severe injuries (severed body, lacerations, bruises and hemorrhaging or major scale loss) [86].

Similar tagging methods involving less invasive means were performed with passive integrated transponder (PIT) tags [87]. Such tags have a small footprint (0.1 g) and do not require battery, as transceiver antennas at installed locations externally energize them. Such

tags have been used to identify paths of preference of migrating fish in hydropower plants and to obtain water allocation routines between bypass and turbines that optimize both fish guidance efficiency and hydroelectric production.

Despite the small signature of PIT systems, radio-telemetry can provide more insight into the survivability of fish that pass through hydropower plants. Such a study involved small devices weighting 4–5 g surgically embedded to the bodies of the examined fish (e.g., eels) [84]. The fish were subsequently released upstream the examined hydropower stations and their status was tracked by stationary or portable receiver antennas placed downstream the examined site. The identification of dead fish was not straightforward, as a stationary transmitter may not necessarily indicate mortality, as the fish may intentionally remain stationary at specific river locations. For this reason, a number of fish were also released downstream the hydropower facility as a control group. This study found that at least 92–96% of the fish passing through the turbine stations were likely to survive; however, it did not find direct turbine mortalities, as it showed that no fish actually passed the turbine. Instead, any detected mortalities were related to injuries at the bypass routes. Alternative to radio signals, which inherently have a small transmission range within water streams, recently developed technologies involve acoustic telemetry. The latter has several advantages over radio telemetry, aside the greater transmission distance; acoustic telemetry transmitters are less invasive, as they do not involve the trailing antenna of the former. In a recent study [88], such a system employed multiple receivers, cabled or autonomous, to determine the 3D fish track by measuring the time of arrival information for the valid detection of embedded tags to fish, revealing the specific route of passage (spillway or turbine) [89].

Aside from the techniques used for tracking fish, the assessment and analysis of the results of such investigations is another topic that requires care. Indeed, it has been recently demonstrated that capturing fish after the hydropower installation and handling them may introduce additional injuries [90]. Specifically, frequently occurring injuries, such as scale loss, tears in the fins, dermal lesions, hemorrhages or bruises could be also induced by subsequent handling after testing, though the intensity of the trauma can be an indication of its origin.

Apart from the aforementioned techniques that involve handling live fish and the associated ethical issues of exposing them intentionally to danger, experimental investigations may involve autonomous sensor devices [91,92] that replicate fish trajectories through a turbine. Such sensor devices, termed as "sensor fish", can provide much more detailed information and analytical data of local accelerations and even pressure and temperature, with sampling rates in the order of 200–2000 Hz. Such devices have been tested in large Francis [93] and Kaplan turbines [94], providing details on the pressure field experienced along the trajectory of "sensor fish", as well as accelerations indicating collisions with blades and structural elements of the turbine. Data from the sensors have been used as pressure input for other studies, see, e.g., [46].

4.2. Assessing Fish Damage—Numerical Methods

4.2.1. Early Attempts

The advancement of computer technology enabled more recent works to be carried out in combination, or in their entirety, with computer models, thus allowing for a much greater insight into hydraulic machinery operation and possible interactions with passing fish. These computer models involve various assumptions, encompassing different degrees of physical mechanisms, complexity and computational cost. One of the first attempts to describe the fish-friendliness of hydraulic turbine systems was to estimate the volume of locations where various metrics relevant to fish damage and mortality, such as pressure, shear or turbulence, as mentioned in Section 3, exceeded specific limits, see [81,95–97]; this volume was subsequently correlated with the possibility of fish mortality. While this straightforward indicator could be employed as a criterion for optimization studies, it can be misleading, since the existence of locations exceeding a particular stressor metric, or its

volume, does not necessarily mean that a fish will pass through this location. Further refinements involved the simplification of treating a fish trajectory as a flow streamline [81,98,99] and deriving pressure or shear stress variation along this path; subsequently, averaging of the calculated values along sampled paths can provide metrics relevant to fish mortality. Still, using streamlines ignores fish inertia and can be misleading.

4.2.2. Particle-Based Methods

More recent investigations involve a Lagrangian perspective for tracking fish motion; in such an approach, a fish is approximated as a particle [100] or some configuration of linked particles with the discrete element method (DEM) [101]. Such Lagrangian, particlebased methods, in any of the forms of the discrete particle method, dense discrete particle method or discrete element method, are very popular in computational fluid dynamics for tracking populations of objects dragged by the flow; hence, they have been investigated for tracking fish motion in hydraulic turbines, see particularly [100–103]. A common characteristic in all Lagrangian methods is that the motion of the tracked bodies is captured by integrating Newton's law of motion, considering various forces that are applicable per case. On the downside, a basic limitation is that in most cases the forces arising from the interaction of the flow with the body are derived based on the assumption of tracking spherical particles. Even if some models may feature corrections for non-spherical shapes, in their core, the three-dimensional body orientation is not taken into account. Moreover, particle-to-particle or particle-to-boundary interactions are also performed assuming point-like objects with no volume, with a notable exception being the discrete element method, which can handle objects approximated as simple, general shapes (e.g., ellipse, cylinder, etc.).

4.2.3. Fully Coupled 6-Degrees-of-Freedom (6-DoF); Overset Meshes

Overset mesh (also known as "Chimera" or "Overlapping" mesh/grid) is a method of manipulating multiple overlaying and disconnected computational domains, by performing interpolations and by enabling/disabling suitably chosen cells, in order to form a single computational continuum over which a flow field can be resolved. The fact that the individual domains are topologically disconnected renders the method highly suitable for the handling of complex motions/deformations [104]. In general, it involves a background mesh and several component meshes, which, in the case of the interest of the present paper, are attached to moving bodies. Thus, during calculations, connectivity between the topologically disconnected meshes is redefined at every time step, whereas the motion of the tracked body is captured accurately, inherently taking into account the volume and exact shape of the tracked body, whereas motion can be described by resolving the 6-DoF equations of motion coupled with the fluid flow. It is a well-established, accurate and versatile method to handle 6-DoF body tracking, indicatively see, e.g., Figure 8, where an accurate fish model was tracked through an axial turbine.

While highly accurate, the main drawbacks of coupled overset 6-DoF are: (a) high computational cost, (b) lack of a default collision detection handling and prevention method, (c) special requirements for constructing the computational meshes, mainly in terms of sizing at the overlap region between component/background meshes and (d) per-case adjustment of schemes for cell-cutting and grid priorities, to reduce the number of cells without suitable interpolation neighbors (also called orphan cells).



Figure 8. Indicative sequence of an fish model, tracked with overset mesh 6-DoF, in an axial flow turbine (for more details refer to [105]). (a) The detailed fish model used to construct the computational mesh of the overset zone (refinement is also visible), (**b**–**e**) overset tracking simulation instances (at 0.42, 0.69, 0.89 and 1.14 s) for an axial turbine.

4.2.4. Fully Coupled 6-Degrees-of-Freedom (6-DoF); Immersed Boundaries

The immersed boundary method (IBM) is a technique of mapping solid geometry on a background mesh [106]; this can be achieved either by altering the flow equations, e.g., through the introduction of appropriate source terms in the momentum equation, or by actually altering the computational mesh through cell-cutting to conform to the body shape. Similar to overset meshes, it is an accurate technique that can describe the precise shape of a tracked body and is rather computationally expensive. Since the method relies on the mapping of the described body to the computational mesh, a natural requirement is that the mesh is fine enough to describe all the necessary details of the body; otherwise, the solid mapping to the fluid domain may fail or will not be accurate (non-smooth surface, artificial gaps/holes). This requirement poses practical resolution restrictions if the tracked body is much smaller than the turbine dimensions, since the required computational mesh will be highly refined and the computational cost intractably high. Recently, the lattice Boltzmann method with immersed boundaries was employed for tracking fish motion through axial turbines [107]; the reported computational cost was in the order of 500–800 h for 1–3 simulated fish. In the author's experience [105] (see also, e.g., Figure 9), immersed boundary techniques can be efficiently applied for bodies roughly 50 times smaller than the turbine; beyond that point, the computational cost becomes prohibitive, unless adaptive refinement is used around the tracked object.



Figure 9. Indicative sequence of an ellipsoid body (the dark blue ellipse), tracked with immersed boundaries, in an axial flow turbine (for more details refer to [105]). Time instances (**a**) 0.45 s, (**b**) 0.87 s and (**c**) 1.21 s.

4.2.5. Uncoupled 6-Degrees-of-Freedom (6-DoF)

A recent variant [108] to the more computationally expensive 6-DoF techniques decouples the flow field from the body tracking. This decoupling makes it possible to solve the flow field with steady-state moving reference frames, which greatly speeds up the solution by roughly two orders of magnitude, allowing for the derivation of statistics (see indicatively Figure 10, where a specifically designed reversible, fish-friendly Deriaz turbine [109] was tested). Since the flow is decoupled from the body motion, the pressure field around the moving body is approximated through the dynamic pressure of the slip velocity. Pressure can be integrated to calculate forces and torques acting on the body. External forces from collisions with rigid boundaries (blades, draft tubes, etc.) can also be taken into account (indicatively, see Figure 11, where an ellipsoidal fish model collides with the rotor of an axial turbine, causing it to tumble).

The major underlying assumption is that the body mass of the tracked object is insignificant compared to the flow momentum; hence, it does not significantly alter the flow field. Numerical experiments have shown that this assumption seems valid when considering fish sizes with characteristic dimensions at most 50 times smaller than a turbine (see [105,108]); indeed, the predicted trajectory with the uncoupled fast tracking is very similar to the fully coupled 6-DoF either with immersed boundaries [105] or overset meshes [108], in the case of axial and diagonal (Deriaz)-type turbines.



Figure 10. Trajectories and observed collisions of fish for different starting positions for a fish-friendly Deriaz-type turbine [109], at turbine mode (**left**) and pumping mode (**right**). Colored spheres indicate the forces above 10 N, corresponding to collisions.



Figure 11. Indicative sequence of an ellipsoid body (the dark blue ellipse), tracked with the fast uncoupled 6-DoF, the fast tracking method. The method also includes collisions, as evidenced at 0.33 s, which causes the tracked body to tumble.

4.2.6. Summary of Computational Methods

Here, a concise qualitative comparison of the different methods is presented in the form of a table, see Table 1. Comparison criteria involve accuracy, computational cost and ability to handle collisions for estimating blade impact.

Table 1. Summary of different methods for tracking fish motion through hydraulic turbines/estimating fish-friendliness of concept designs. Plus sign (+) indicates advantage, minus sign (-) indicates disadvantage.

Method	Accuracy	Computational Cost	Collision Detection
Volume-based criteria	_	++++	-
Streamlines	++	++++	-
Particle-based	+++	+	+
Fully coupled-overset	+++++	-	-/(not by default $)$
Fully coupled immersed boundary	+++++	-	-/(not by default $)$
Uncoupled 6-DoF	++++	+++	+

5. Derivation of Indices/Metrics to Assess the Fish-Friendliness of Hydro-Turbines

The aforementioned methods, either experimental correlations or numerical techniques, are helpful to quantify the severity of different stressor factors that affect fish survivability. However, the combinatory effect of all these stressors has to be integrated into a metric that can be used in the design and planning phase of hydropower infrastructure. One of the first attempts involved only blade-strike estimations to estimate the probability of such events [110], based on the fish size, the turbine type and the operating conditions of the turbine.

Subsequent iterations involved refinements and considerations of additional factors aside blade strike. Such a metric, developed by the Pacific Northwest National Laboratory, is the biological performance assessment (BioPA) method [19]; this method estimates the probabilities that fish will encounter specific conditions during passage through a hydraulic turbine. This metric also relies on specific dose–response relationships between species

of fish and known injury mechanisms, which are determined by subjecting a suitable number of fish to various magnitudes of a stressor (see Section 3) and computing the probability of injury or mortality at each magnitude. BioPA employs four metrics for assessing mortality: nadir pressure, shear, turbulence and blade strike, which are integrated over time, to estimate the BioPA score for each individual stressor factor. Indicatively, mortality probability, *P*_{mort}, due to pressure is estimated through the following relation [25]:

$$P_{mort} = \frac{e^{-5.56 + 3.85 \cdot LRP_i}}{1 + e^{-5.56 + 3.85 \cdot LRP_i}} \tag{1}$$

where *LRP* stands for the natural log of the pressure change ratio of acclimation pressure p_A , to nadir (i.e., lowest possible) pressure, p_L :

$$LRP_i = \ln\left(\frac{p_A}{p_L}\right) \tag{2}$$

The overall risk of mortal injury during passage is evaluated with a performance score, β , accounting for the probability, $P(p_L)$, of exposure at different minimum pressure levels, by sampling all the examined streamlines:

$$\beta = \left(1 - \sum_{i=1}^{N} P_{mort,i} \cdot P(p_L)_i\right) \cdot 100\%$$
(3)

The authors of the method highlight that Equation (1) does not describe delayed mortality effects. For blade strike, the probability P_{blade} is estimated through the following equation [111]:

$$P_{blade} = \frac{nNL\cos(\theta)}{60V_{ax}} \cdot 100\%$$
(4)

where *n* stands for runner rpm, *N* is the number of blades, *L* is the fish length, θ is the angle between absolute and axial (or radial) velocity vector and *V* stands for the velocity, either axial or radial, depending on the application (axial/radial turbine). In any case, various stressor scores are combined into a single operating condition score using a weighting algorithm. Statistical sampling is used for fish tracks, approximated as streamlines, i.e., neutrally buoyant particle traces, which are emitted uniformly from the turbine inlet section (unless site-specific information is available and suggested otherwise). Inherently, this method requires a numerical solution of the flow field in the turbine, using computational fluid dynamics, to provide the stressor values and the streamlines. This metric has been used in several studies, see [25,76], with the aim to expand its scope to more complex physical mechanisms, such as considering fish inertia and turbulence effects.

Another, more recent, metric is the European fish hazard index (EFHI) [17], which integrates species-specific sensitivities of the ambient fish community, derived from species' life-history traits and conservation value, and the specific operational, constructional and technical characteristics of a hydropower plant site. The input information for this index is: (1) turbine dimensions, type and operating conditions; (2) fish migration or protection facilities; (3) target species and (4) stream reach. These categories are then cross-tabulated and contrasted to the rounded integer value of the species' biological sensitivity to produce a numerical score for each hazard and species. The EFHI heavily relies on semi-empirical data, relations and regressions, to derive these scores; for example, blade strike mortality, denoted as M_{Monten} , is estimated as:

$$M_{Monten} = \frac{0.5L_{max}}{S_{rel}} \cdot 100\%$$
⁽⁵⁾

where L_{max} is the ratio of intake screen gap-width to a geometric parameter, *b*, dependent on the fish species (e.g., 0.11 for fusiform fish, or 0.03 for eel-like body shapes) and S_{rel} is the relative space between blades, taking into account the blade angle, annular passage

diameters and number of blades. Barotrauma risk is simply related to the barrier/dam height and hence classified as low for H < 2 m, moderate for 2–10 m and high for >10 m; subsequently, it is adjusted to the particularities of each examined fish species (in particular, the structure and presence of their gas bladder, e.g., physostomous or physoclistous). In any case, the entire process is performed within an Excel worksheet, which, based on the aforementioned user input, determines individual hazard scores with values of 0, 0.25, 0.5, 0.75 or 1, with higher scores indicating a more severe hazard. The implementation as is can involve ensemble estimations for up to five target species that are considered by the user to best reflect the local fish population, conservation status and river region. In case of specific conservational status or environmental concern, selected species can be individually categorized to the highest sensitivity class. The final EFHI score is finally calculated as an aggregation of all hazard scores.

A general observation, based on the aforementioned details of the BioPA and EFHI, is that BioPA relies more on detailed flow field information data of a particular turbine design (i.e., manipulations of a 3D flow field with streamlines, etc.), whereas the EFHI relies on regressions and empirical data entirely. It is thus expected that the EFHI may be more successful in assessing traditional turbine concepts and existing installations/refurbished sites, for which the assessment criteria fall within the calibration range of the regressions/models used. On the other hand, BioPa, even though more detailed, is expected to be much more computationally intensive; hence, seems suitable to designing new turbine concepts.

6. Fish-Friendly Hydropower Design Concepts

The US Environment Protection Agency has established specific considerations and engineering-based criteria that are considered critical in the design of fish-friendly turbines; these criteria involve [16,44]:

- Blunt, thick blades (see indicatively the Restoration Hydro Turbine or RHT in [71], or Figure 12a);
- Low rotational speed of the turbine runner;
- Large passages;
- Few blades;
- No exposed gaps.

However, these criteria mainly address the blade/fish collision, rather than consist in a holistic approach in assessing all potential mechanisms of fish damage. In any case, when considering existing designs, field studies indicate that Francis turbines resulted in a higher immediate mortality risk than Kaplan turbines relative to controls [16].

In recent years, several new turbine concepts, or modifications/variations of existing concepts, have emerged, specifically to address fish-friendly passage. One such example is the minimum gap runner (MGR) turbine [112], see Figure 12b; this design is very similar to the Kaplan turbine, but the blades, the hub and the discharge ring have a more spherical profile to reduce clearances; hence, it was expected that the performance against mechanical injury would be improved. This design modification also removes part of the leading edge and, as a result, also reduces the local pressure gradients and shear stresses, while also having a positive effect in the efficiency of the turbine [113]. Indeed, the testing of the MGR turbine showed improvement in fish injuries compared to a traditional Kaplan turbine, with only 1.4% of the fish being injured, compared to 2.5% of the statistical control group tested with a Kaplan turbine [114].

Similar to the MGR, the very low head (VLH) turbine can be considered a Kaplan-type turbine, specifically engineered for safe fish passage [115]. This turbine was originally developed in France, to be installed at an existing low head hydropower facility, for heads of 1.4–4.2 m and flows between 10 and 30 m³/s, mounted in an inclined position (30–50°) from horizontal and using an eight-bladed Kaplan-style runner, as seen in Figure 12c. The unit is designed to be compact and easily integrated to existing facilities, by reducing intake and outlet structure sizes by maximizing the diameter of the turbine runner. The large diameter of the turbine contributes to slow rotational speeds, which is a major requirement

for safe fish passage. Indeed, testing in Europe indicated mortality rates of less than 5% [115]; however, similar projects in North America have been hampered by extremely cold temperatures and ice.



Figure 12. Fish-friendly turbine concepts: (**a**) Natel energy blunt, thick-blade runner, (**b**) minimum gap runner (MGR) turbine, (**c**) very low head (VLH) turbine, (**d**) vaneless swirl injector Kaplan turbine, (**e**) Alden turbine, (**f**) Archimedes turbine, (**g**) vortex turbine, (**h**) hydrokinetic hub-less turbine, (**i**) Darrieus-type hydrokinetic turbine.

Another Kaplan turbine design variation is the vaneless swirl injector system [116], seen in Figure 12d; in this design, the traditional inlet guide vane (IGV) array is replaced with a vaneless casing, which is coupled with a fish-friendly Kaplan turbine and is applicable for a head up to 6 m. The vaneless casing creates the required swirl velocity component necessary for the runner by guiding the flow through a spiral-shaped track. Since there are no vanes, fish mortality caused by strike damage is dramatically reduced.

An innovative turbine concept that was developed at the same time as the MGR turbine is the Alden turbine [112,117], seen in Figure 12e. It is a low head turbine (H~25 m) developed jointly by Alden Research Laboratory, Inc./Northern Research and Engineering Corporation (ARL/NREC) and Voith Hydro, Inc. (Voith) in 1995. The final proposed design has three helical blades, which are attached both to the hub and to an external rotating shroud, hence eliminating gaps and the possibility of grinding injury. The casing of the turbine has a traditional scroll/spiral shape, directing the flow through a radial space with few but long vane blades (also termed as wicket gates) and then through a

gradual downturn toward the runner entrance. The long helical shape of the rotor blades was specifically designed to reduce pressure changes and to improve fish survival due to barotrauma through the turbine. Furthermore, the blades of the Aden turbine feature a thick leading edge, which has been demonstrated to reduce blunt body impact injury. Thus, these design choices in the Alden turbine concept, namely the reduction in the number of blades, the reduction in the operating speed, the minimization of pressure differences and the thick leading edges, have been demonstrated to greatly improve fish survivability.

The Archimedes turbine [118,119], seen in Figure 12f, although not new, is a very simple design, well suited for low head installations (H < 10 m). The Archimedes turbine consists of a screw-shaped runner, placed within a co-axial, tubular shroud. When water enters the top of the shaft, the weight of the water pushes on the screw blades, causing the shaft to rotate and allowing for the water to fall to the lower level. Such designs are intended for low heads and have diameters between 1.5 and 3.5 m. Their low rotational speeds, minimal pressure changes and shear forces minimize injury to fish that pass through this turbine. The Archimedes screw turbine particularly as a fish-friendly solution has been installed and tested at several sites in Europe and the UK with positive results [120–122]. Studies indicate minimal to no injuries and no mortalities from the downstream passage of adult European eels, larval and juvenile river lamprey, sea-run Brown Trout and Atlantic Salmon kelts.

The vortex turbine design Is installed as a by-pass system to a dam or classical weir, and which is intended to permit two-way fish migration [123], as seen in Figure 12g. It consists of a vortex pool where a strong free-surface vortex develops, in the center of which a Francis-like turbine is located. The water leaves the pool through an orifice at the bottom and flows back into the river through an outlet channel. The turbine is inherently designed to be operated as a low-head device (as low as H~0.7 m) [124], operating at low rotational speeds (indicatively, 30 rpm). The rotating speed of the runner makes it possible to achieve a very small slip tangential velocity, so there is a low risk of fish/blade impact. Additionally, clearances can be adjusted, to allow for fish to pass freely in both ways.

Entirely alternative to reservoir-based, or diversion-based hydropower discussed so far, are in-stream turbines, which can be placed directly in the water stream, thus relying heavily on the flow kinetic energy of the incoming water. A version of this concept, seen in Figure 12h, was recently proposed [125], which resembles a hub-less axial turbine. This concept was found to satisfy pressure drop and strain rate criteria [125], conforming to requirements for safe fish passage [74]. However, the authors of the work identify that the design is an extremely low head turbine and has a low efficiency (in fact calculated to be much lower than the Betz limit). Similar concepts of in-stream turbines have been examined for vertical-axis turbines, such as Darrieus-type [126] turbines, as seen in Figure 12i, which are found to have an efficiency of up to 50%, whereas the estimated survivability through them is of more than 98% [127].

Finally, some additional factors, even if not directly related to turbine technologies/subsystems, still affect fish mortality [128]:

- Oxygen deficiency in the water of reservoirs of hydropower plants, which can cause damage to downstream ecosystems. Hence, artificial aeration of the flowing water is a consideration that needs to be taken into account for successful fish-friendly turbines [129];
- Oil leaks; oil is used both as a means of lubrication and as a means of actuation of turbine blades (e.g., in the case of Kaplan turbines). To date, several Kaplan–Bulb and Francis turbines have been upgraded so as to make them work free from oil, and new materials and lubricants are being developed.

7. Conclusions and Future Perspectives

It is clear that hydropower will continue to play an important role in energy production worldwide; thus, ecological pressure to aquatic life is expected to increase. The necessity to design new concepts that holistically approach aquatic life, either as a mean of deterrence, i.e., leading migrating fish through a bypass stream, or protecting passing fish through the turbine, is becoming apparent.

While there are guidelines and estimates for the damage thresholds of various fish species, useful to predict potential damage as they pass through a turbine, these thresholds vary wildly and are strongly dependent on the exact specie examined. This fact, along with the inherent site-specific nature of hydropower sites, renders estimations of survivability unique for each case and difficult to generalize easily. Some success was found with semi-empirical models; however, the progressive integration of computational fluid dynamics (CFD) in the design phase of hydropower can provide much more insight than any experimental technique. Moreover, CFD predictions can be combined with damage thresholds to assess the survivability of a candidate design.

Indeed, CFD predictions already have contributed greatly to the design of existing fishfriendly concepts. However, all these concepts are applicable only to low head applications (<40 m). A future challenge is clearly the development of such fish-friendly concepts at higher operating heads and other turbine types (e.g., Francis and Pelton).

Author Contributions: Conceptualization, P.K. and J.A.; methodology, P.K.; formal analysis, P.K.; investigation, P.K. and J.A.; resources, P.K. and J.A.; data curation, P.K. and J.A.; writing—original draft preparation, P.K.; writing—review and editing, P.K. and J.A.; visualization, P.K.; supervision, J.A.; project administration, J.A.; funding acquisition, P.K. All authors have read and agreed to the published version of the manuscript.

Funding: This research has been co-financed by the European Union and Greek national funds through the Operational Program Competitiveness, Entrepreneurship & Innovation, under the call RESEARCH–CRE-ATE–INNOVATE (T1EDK-01334, 2018–21).



Data Availability Statement: Not applicable.

Conflicts of Interest: The authors declare no conflict of interest.

References

- 1. BP p.l.c. Bp Statistical Review of World Energy; BP Plc: London, UK, 2022.
- 2. IRENA. Renewable Energy Statistics 2022; International Renewable Energy Agency: Abu Dhabi, United Arab Emirates, 2022.
- 3. International Energy Agency. Hydropower; International Energy Agency: Paris, France, 2021.
- 4. Jager, H.I.; De Silva, T.; Uria-Martinez, R.; Pracheil, B.M.; Macknick, J. Shifts in Hydropower Operation to Balance Wind and Solar Will Modify Effects on Aquatic Biota. *Water Biol. Secur.* 2022, *1*, 100060. [CrossRef]
- 5. Murdock, H.; Duncan, G.; Thomas, A. Renewables Global Status Report; International Nuclear Information System: Vienna, Austria, 2021.
- 6. International Energy Agency. *How Rapidly Will the Global Electricity Storage Market Grow by* 2026? International Energy Agency: Paris, France, 2021.
- Killingtveit, Å. 15—Hydroelectric Power; Elsevier Ltd.: Amsterdam, The Netherlands, 2020; ISBN 9780081028865.
- Breeze, P. The Hydropower Resource, Hydropower Sites and Types of Hydropower Plants. In *Hydropower*; Elsevier: Amsterdam, The Netherlands, 2018; pp. 13–21.
- 9. World Energy Council. World Energy Resources, 2013 Survey; World Energy Council: London, UK, 2013.
- 10. Zwahlen, R. *Assessing the Environmental Impacts of Hydropower Projects;* Environmental Earth Sciences; Springer International Publishing: Cham, Switzerland, 2022; ISBN 978-3-030-91184-3.
- Reid, A.J.; Carlson, A.K.; Creed, I.F.; Eliason, E.J.; Gell, P.A.; Johnson, P.T.J.; Kidd, K.A.; MacCormack, T.J.; Olden, J.D.; Ormerod, S.J.; et al. Emerging Threats and Persistent Conservation Challenges for Freshwater Biodiversity. *Biol. Rev.* 2019, 94, 849–873. [CrossRef] [PubMed]
- Dudgeon, D.; Arthington, A.H.; Gessner, M.O.; Kawabata, Z.-I.; Knowler, D.J.; Lévêque, C.; Naiman, R.J.; Prieur-Richard, A.-H.; Soto, D.; Stiassny, M.L.J.; et al. Freshwater Biodiversity: Importance, Threats, Status and Conservation Challenges. *Biol. Rev.* 2006, *81*, 163. [CrossRef]
- Twardek, W.M.; Cowx, I.G.; Lapointe, N.W.R.; Paukert, C.; Beard, T.D.; Bennett, E.M.; Browne, D.; Carlson, A.K.; Clarke, K.D.; Hogan, Z.; et al. Bright Spots for Inland Fish and Fisheries to Guide Future Hydropower Development. *Water Biol. Secur.* 2022, 1, 100009. [CrossRef]

- 14. Almond, R.E.A.; Grooten, M.; Petersen, T. Living Planet Report 2020—Bending the Curve of Biodiversity Loss; World Wildlife Fund: Gland, Switzerland, 2020.
- 15. Radinger, J.; van Treeck, R.; Wolter, C. Evident but Context-dependent Mortality of Fish Passing Hydroelectric Turbines. *Conserv. Biol.* **2022**, *36*, e13870. [CrossRef]
- Algera, D.A.; Rytwinski, T.; Taylor, J.J.; Bennett, J.R.; Smokorowski, K.E.; Harrison, P.M.; Clarke, K.D.; Enders, E.C.; Power, M.; Bevelhimer, M.S.; et al. What Are the Relative Risks of Mortality and Injury for Fish during Downstream Passage at Hydroelectric Dams in Temperate Regions? A Systematic Review. *Environ. Evid.* 2020, *9*, 3. [CrossRef]
- 17. Van Treeck, R.; Radinger, J.; Smialek, N.; Pander, J.; Geist, J.; Mueller, M.; Wolter, C. Comparative Assessment of Hydropower Risks for Fishes Using the Novel European Fish Hazard Index. *Sustain. Energy Technol. Assess.* **2022**, *51*, 101906. [CrossRef]
- Geiger, F.; Stoltz, U. Guidelines for Application of Different Analysis Methods of Fish Passage Through Turbines—Impact Assessment of Fish Behavioural Aspects. In *Novel Developments for Sustainable Hydropower*; Springer International Publishing: Cham, Switzerland, 2022; pp. 105–115.
- Richmond, M.C.; Serkowski, J.A.; Rakowski, C.L.; Strickler, B.; Weisbeck, M.; Dotson, C.L. Computational Tools to Assess Turbine Biological Performance. *Hydro Rev.* 2014, 33, 88–97.
- Knott, J.; Mueller, M.; Pander, J.; Geist, J. Bigger than Expected: Species- and Size-Specific Passage of Fish through Hydropower Screens. Ecol. Eng. 2023, 188, 106883. [CrossRef]
- Wolter, C.; Bernotat, D.; Gessner, J.; Brüning, A.; Lackemann, J.; Radinger, J. Fachplanerische Bewertung Der Mortalität von Fischen an Wasserkraftanlagen. Leibniz-Institut für Gewässerökologie und Binnenfischerei: Berlin, Germany, 2000; Volume 561, ISBN 9783896242990.
- 22. Ploskey, G.R.; Carlson, T.J. *Comparison of Blade Strike Modeling Results with Empirical Data*; EERE Publication and Product Library: Richland, WA, USA, 2004.
- 23. Dadswell, M. Macrotidal Estuaries: A Region of Collision between Migratory Marine Animals and Tidal Power Development. *Biol. J. Linn. Soc.* **1994**, *51*, 93–113. [CrossRef]
- Coutant, C.C.; Whitney, R.R. Fish Behavior in Relation to Passage through Hydropower Turbines: A Review. *Trans. Am. Fish. Soc.* 2000, 129, 351–380. [CrossRef]
- 25. Richmond, M.C.; Romero-Gomez, P.; Serkowski, J.A.; Rakowski, C.L.; Graf, M.J. Comparative Study of Barotrauma Risk during Fish Passage through Kaplan Turbines; Pacific Northwest National Lab.: Richland, WA, USA, 2015.
- IEA Hydropower Annex XVI—Hidden Hydro. Available online: https://www.ieahydro.org/work-programme/annex-xvihidden-hydro (accessed on 11 January 2023).
- Hydropower Europe What Research and Innovation are Needed to Tap More Hidden Hydro Opportunities in the Future? Available online: https://hydropower-europe.eu/latest-news/webinar-28-february-2022-what-research-and-innovation-areneeded-to-tap-more-hidden-hydro-opportunities-in-the-future/ (accessed on 11 January 2023).
- Quaranta, E.; Bódis, K.; Kasiulis, E.; McNabola, A.; Pistocchi, A. Is There a Residual and Hidden Potential for Small and Micro Hydropower in Europe? A Screening-Level Regional Assessment. *Water Resour. Manag.* 2022, 36, 1745–1762. [CrossRef]
- European Commision Development of Hydropower Equipment for Hidden Hydropower, TOPIC ID: HORIZON-CL5-2021-D3-03-11. Available online: https://ec.europa.eu/info/funding-tenders/opportunities/portal/screen/opportunities/topic-details/ horizon-cl5-2021-d3-03-11 (accessed on 11 January 2023).
- 30. Schmutz, S.; Sendzimir, J. *Riverine Ecosystem Management*; Schmutz, S., Sendzimir, J., Eds.; Springer International Publishing: Cham, Switzerland, 2018; ISBN 978-3-319-73249-7.
- 31. Gøtske, E.K.; Victoria, M. Future Operation of Hydropower in Europe under High Renewable Penetration and Climate Change. *iScience* **2021**, 24, 102999. [CrossRef]
- 32. Kougias, I.; Aggidis, G.; Avellan, F.; Deniz, S.; Lundin, U.; Moro, A.; Muntean, S.; Novara, D.; Pérez-Díaz, J.I.; Quaranta, E.; et al. Analysis of Emerging Technologies in the Hydropower Sector. *Renew. Sustain. Energy Rev.* **2019**, *113*, 109257. [CrossRef]
- Yang, C.; Zheng, Y.; Zhang, Y.; Luo, H. A Review of Research on the Design of Fish-Friendly Hydraulic Turbines. *Chin. J. Eng. Sci.* 2018, 20, 96. [CrossRef]
- 34. Pereira, G.M. Design of Hydroelectric Power Plants—Step by Step; CRC Press: Boca Raton, FL, USA, 2021; ISBN 978-0367751722.
- 35. Kim, J.; Mandrak, N.E. Effects of a Vertical Electric Barrier on the Behaviour of Rainbow Trout. *Aquat. Ecosyst. Health Manag.* 2019, 22, 183–192. [CrossRef]
- Tutzer, R.; Röck, S.; Walde, J.; Haug, J.; Brinkmeier, B.; Aufleger, M.; Unfer, G.; Führer, S.; Zeiringer, B. A Physical and Behavioral Barrier for Enhancing Fish Downstream Migration at Hydropower Dams: The Flexible Fish Protector. *Water* 2022, 14, 378. [CrossRef]
- Tulio, C.; Faria, M.; Maria, F.; Viana, E.; Martinez, C.B. Model Development of a Fish Electromechanical Barrier for Hydraulic Turbines. Int. J. Ecosyst. 2014, 4, 159–164. [CrossRef]
- 38. Larinier, M. Fish Passage Experience at Small-Scale Hydro-Electric Power Plants in France. Hydrobiologia 2008, 609, 97–108. [CrossRef]
- US DOE Office of Energy Efficiency and Renewables Types of Hydropower Plants. Available online: https://www.energy.gov/ eere/water/types-hydropower-plants (accessed on 14 January 2023).

- Wang, L.; Cui, J.; Shu, L.; Jiang, D.; Xiang, C.; Li, L.; Zhou, P. Research on the Vortex Rope Control Techniques in Draft Tube of Francis Turbines. *Energies* 2022, 15, 9280. [CrossRef]
- 41. Intenational Renewable Energy Agency (IRENA). *Innovation Landscape Brief: Innovative Operation of Pumped Hydropower Storage;* IRENA: Abu Dhabi, United Arab Emirates, 2020.
- 42. Sale, M.J.; Cada, G.F.; Carlson, T.J.; Dauble, D.D.; Hunt, R.T.; Sommers, G.L.; Rinehart, B.N.; Flynn, J.V.; Brookshier, P.A. *DOE Hydropower Program Annual Report for FY 2001*; EERE Publication and Product Library: Washington, DC, USA, 2002.
- 43. Cada, G.F.; Coutant, C.C.; Whitney, R.R.; Washington, L. *Development of Biological Criteria for the Design of Advanced Hydropower Turbines*; EERE Publication and Product Library: Washington, DC, USA, 1997.
- 44. Cooke, S.J.; Hatry, C.; Hasler, C.T.; Smokorowksi, K.E. *Literature Review, Synthesis and Proposed Guidelines Related to the Biological Evaluation of "Fish Friendly" Very Low Head Turbine Technology*; Department of Fisheries and Oceans: Ottawa, ON, Canada, 2011.
- 45. Sale, M.J.; Cada, G.F.; Carlson, T.J.; Hunt, R.T.; Sommers, G.L.; Rinehart, B.N. DOE Hydropower Program Annual Report for FY 2002, DOE/ID-11107; Idaho National Lab.: Idaho Falls, ID, USA, 2003.
- 46. Stephenson, J.R.; Gingerich, A.J.; Brown, R.S.; Pflugrath, B.D.; Deng, Z.; Carlson, T.J.; Langeslay, M.J.; Ahmann, M.L.; Johnson, R.L.; Seaburg, A.G. Assessing Barotrauma in Neutrally and Negatively Buoyant Juvenile Salmonids Exposed to Simulated Hydro-Turbine Passage Using a Mobile Aquatic Barotrauma Laboratory. *Fish. Res.* 2010, *106*, 271–278. [CrossRef]
- Guo, T.; Zhang, J.; Luo, Z. Analysis of Channel Vortex and Cavitation Performance of the Francis Turbine under Partial Flow Conditions. *Processes* 2021, 9, 1385. [CrossRef]
- 48. Kumar, P.; Saini, R.P. Study of Cavitation in Hydro Turbines—A Review. Renew. Sustain. Energy Rev. 2010, 14, 374–383. [CrossRef]
- 49. Brijkishore; Khare, R.; Prasad, V. Prediction of Cavitation and Its Mitigation Techniques in Hydraulic Turbines—A Review. *Ocean Eng.* **2021**, *221*, 108512. [CrossRef]
- 50. Zhu, L.; Zhang, R.; Yu, A.; Lu, L.; Luo, X. Suppression of Vortex Rope Oscillation and Pressure Vibrations in Francis Turbine Draft Tube Using Various Strategies. *J. Hydrodyn.* **2021**, *33*, 534–545. [CrossRef]
- Rajan, G.K.; Cimbala, J.M. Computational and Theoretical Analyses of the Precessing Vortex Rope in a Simplified Draft Tube of a Scaled Model of a Francis Turbine. J. Fluids Eng. 2017, 139, 021102. [CrossRef]
- 52. Goyal, R.; Cervantes, M.J.; Gandhi, B.K. Vortex Rope Formation in a High Head Model Francis Turbine. J. Fluids Eng. 2017, 139, 041102. [CrossRef]
- 53. Kumar, S.; Cervantes, M.J.; Gandhi, B.K. Rotating Vortex Rope Formation and Mitigation in Draft Tube of Hydro Turbines—A Review from Experimental Perspective. *Renew. Sustain. Energy Rev.* **2021**, *136*, 110354. [CrossRef]
- Tsvetkov, V.I.; Pavlov, D.S.; Nezdoliy, V.K. Changes of Hydrostatic Pressure Lethal to Young of Some Freshwater Fish. J. Ichthyol. 1972, 12, 307–318.
- 55. Morgan, R.P.; Ulanowicz, R.E.; Rasin, V.J.; Noe, L.A.; Gray, G.B. Effects of Shear on Eggs and Larvae of Striped Bass, Morone Saxatilis, and White Perch, M. Americana. *Trans. Am. Fish. Soc.* **1976**, *105*, 149–154. [CrossRef]
- 56. Lampert, W. Experiments on the Resistance of Fish to Rapid Increase in Hydrostatic Pressure. J. Fish Biol. 1976, 8, 381–383. [CrossRef]
- 57. Harvey, H.H. Pressure in the Early Life History of Sockeye Salmon. Ph.D. Thesis, University of British Columbia, Vancouver, BC, Canada, 1963.
- 58. Foye, R.E.; Scott, M. Effects of Pressure on Survival of Six Species of Fish. Trans. Am. Fish. Soc. 1965, 94, 88–91. [CrossRef]
- 59. Feathers, M.G.; Knable, A.E. Effects of Depressurization Upon Largemouth Bass. North Am. J. Fish. Manag. 1983, 3, 86–90. [CrossRef]
- Brown, R.S.; Pflugrath, B.D.; Colotelo, A.H.; Brauner, C.J.; Carlson, T.J.; Deng, Z.D.; Seaburg, A.G. Pathways of Barotrauma in Juvenile Salmonids Exposed to Simulated Hydroturbine Passage: Boyle 's Law vs. Henry 's Law. *Fish. Res.* 2012, 121–122, 43–50. [CrossRef]
- Brown, R.S.; Carlson, T.J.; Welch, A.E.; Stephenson, J.R.; Abernethy, C.S.; Ebberts, B.D.; Langeslay, M.J.; Ahmann, M.L.; Feil, D.H.; Skalski, J.R.; et al. Assessment of Barotrauma from Rapid Decompression of Depth-Acclimated Juvenile Chinook Salmon Bearing Radiotelemetry Transmitters. *Trans. Am. Fish. Soc.* 2009, 138, 1285–1301. [CrossRef]
- 62. Franc, J.-P.; Michel, J.-M. Fundamentals of Cavitation; Kluwer Academic Publishers: Dordrecht, The Netherlands, 2005.
- 63. Krizek, J.; Delrot, P.; Moser, C. Repetitive Regime of Highly Focused Liquid Microjets for Needle-Free Injection. *Sci. Rep.* 2020, 10, 5067. [CrossRef]
- Reuter, F.; Deiter, C.; Ohl, C.-D. Cavitation Erosion by Shockwave Self-Focusing of a Single Bubble. Ultrason. Sonochem. 2022, 90, 106131. [CrossRef]
- 65. Koukouvinis, P.; Bruecker, C.; Gavaises, M. Unveiling the Physical Mechanism behind Pistol Shrimp Cavitation. *Sci. Rep.* **2017**, 7, 13994. [CrossRef]
- 66. Muir, J.F. Passage of Young Fish Through Turbines. J. Power Div. 1959, 85, 23–46. [CrossRef]
- van Esch, B.P.M.; Spierts, I.L.Y. Validation of a Model to Predict Fish Passage Mortality in Pumping Stations. *Can. J. Fish. Aquat. Sci.* 2014, 71, 1910–1923. [CrossRef]
- Vikström, L.; Leonardsson, K.; Leander, J.; Shry, S.; Calles, O.; Hellström, G. Validation of Francis–Kaplan Turbine Blade Strike Models for Adult and Juvenile Atlantic Salmon (*Salmo salar*, L.) and Anadromous Brown Trout (*Salmo trutta*, L.) Passing High Head Turbines. *Sustainability* 2020, 12, 6384. [CrossRef]

- 69. Deng, Z.; Carlson, T.J.; Ploskey, G.R.; Richmond, M.C.; Dauble, D.D. Evaluation of Blade-Strike Models for Estimating the Biological Performance of Kaplan Turbines. *Ecol. Model.* **2007**, *208*, 165–176. [CrossRef]
- 70. Turnpenny, A.W.; Davis, M.H.; Fleming, J.M.; Davies, J.K. *Experimental Studies Relating to the Passage of Fish and Shrimps Through Tidal Power Turbines*; University of Massachusetts: Amherst, MA, USA, 1992.
- Amaral, S.V.; Watson, S.M.; Schneider, A.D.; Rackovan, J.; Baumgartner, A. Improving Survival: Injury and Mortality of Fish Struck by Blades with Slanted, Blunt Leading Edges. J. Ecohydraulics 2020, 5, 175–183. [CrossRef]
- 72. Cook, T.; Hecker, G.; Faulkner, H.; Jiansem, W. Development of a More Fish Tolerant Turbine Runner. In *Technical Memorandum* #2: Development of Biological Design Criteria, DOE/ID-10571; Alden Research Lab.: Holden, MA, USA, 1997.
- 73. Colotelo, A.; Cooke, S.; Smokorowski, K. Application of Forensic Techniques to Enhance Fish Conservation and Management: Injury Detection Using Presumptive Tests for Blood. *Endanger. Species Res.* **2009**, *9*, 169–178. [CrossRef]
- 74. Odeh, M. *A Summary of Environmentally Friendly Turbine Design Concepts;* US Department of Energy Idaho Operations Office: Idaho Falls, ID, USA, 1999.
- 75. Harding, S.F.; Mueller, R.P.; Richmond, M.C.; Romero-Gomez, P.; Colotelo, A.H. Fish Response to Turbulence Generated Using Multiple Randomly Actuated Synthetic Jet Arrays. *Water* **2019**, *11*, 1715. [CrossRef]
- Pflugrath, B.D.; Saylor, R.K.; Engbrecht, K.; Mueller, R.P.; Stephenson, J.R.; Bevelhimer, M.; Pracheil, B.M.; Colotelo, A.H. Biological Response Models: Predicting Injury and Mortality of Fish During Downstream Passage through Hydropower Facilities; Pacific Northwest National Lab. (PNNL): Richland, WA, USA, 2020.
- Čada, G.; Loar, J.; Garrison, L.; Fisher, R.; Neitzel, D. Efforts to Reduce Mortality to Hydroelectric Turbine-Passed Fish: Locating and Quantifying Damaging Shear Stresses. *Environ. Manag.* 2006, 37, 898–906. [CrossRef]
- 78. Cada, G.F. Determining the Effect of Shear Stress on Fish Mortality during Turbine Passage. Hydro Rev. 2007, 26.
- 79. Groves, A.B. *Effects of Hydraulic Shearing Actions on Juvenile Salmon: Summary Report;* National Marine Fisheries Service: Seattle, WA, USA, 1972.
- 80. Johnson, R.L. *Fingerling Fish Research, High-Velocity Flow Through Four-Inch Nozzle;* U.S. Army Corps of Engineers: Portland, OR, USA, 1972.
- Garrison, L.A.; Fisher, R.K.; Sale, M.J.; Cada, G.F. Application of Biological Design Criteria and Computational Fluid Dynamics to Investigate Fish Survival in Kaplan Turbines. In *HydroVision 2002 Technical Papers*; HCI Publications: Kansas City, MO, USA, 2002.
- 82. Cada, G.F. Shaken, Not Stirred: The Recipe for a Fish-Friendly Turbine. Proc. Int. Conf. Hydropower—Waterpower 1997, 1, 374–382.
- Killgore, K.J.; Miller, A.C.; Conley, K.C. Effects of Turbulence on Yolk-Sac Larvae of Paddlefish. Trans. Am. Fish. Soc. 1987, 116, 670–673. [CrossRef]
- Økland, F.; Teichert, M.A.K.; Havn, T.T.B.; Thorstad, E.B.E.; Heermann, L.; Saether, S.A.; Tambets, M.; Borcherding, J.; Sæther, S.A.; Tambets, M.; et al. *Downstream Migration of European Eel at Three German Hydropower Stations*; Norwegian Institute for Nature Research (NINA): Trondheim, Norway, 2017; ISBN 9788242630667.
- 85. Heisey, P.G.; Mathur, D.; Rineer, T. A Reliable Tag–Recapture Technique For Estimating Turbine Passage Survival: Application to Young-of-the-Year American Shad (*Alosa sapidissima*). *Can. J. Fish. Aquat. Sci.* **1992**, *49*, 1826–1834. [CrossRef]
- 86. Mathur, D.; Heisey, P.G.; Euston, E.T.; Skalski, J.R.; Hays, S. Turbine Passage Survival Estimation for Chinook Salmon Smolts (*Oncorhynchus tshawytscha*) at a Large Dam on the Columbia River. *Can. J. Fish. Aquat. Sci.* **1996**, *53*, 542–549. [CrossRef]
- Haraldstad, T.; Höglund, E.; Kroglund, F.; Haugen, T.O.; Forseth, T. Common Mechanisms for Guidance Efficiency of Descending Atlantic Salmon Smolts in Small and Large Hydroelectric Power Plants. *River Res. Appl.* 2018, 34, 1179–1185. [CrossRef]
- McMichael, G.A.; Eppard, M.B.; Carlson, T.J.; Carter, J.A.; Ebberts, B.D.; Brown, R.S.; Weiland, M.; Ploskey, G.R.; Harnish, R.A.; Deng, Z.D. The Juvenile Salmon Acoustic Telemetry System: A New Tool. *Fisheries* 2010, 35, 9–22. [CrossRef]
- Martinez, J.; Fu, T.; Li, X.; Hou, H.; Wang, J.; Eppard, M.B.; Deng, Z.D. A Large Dataset of Detection and Submeter-Accurate 3-D Trajectories of Juvenile Chinook Salmon. *Sci. Data* 2021, *8*, 211. [CrossRef]
- 90. Mueller, M.; Pander, J.; Geist, J. Evaluation of External Fish Injury Caused by Hydropower Plants Based on a Novel Field-Based Protocol. *Fish. Manag. Ecol.* 2017, 24, 240–255. [CrossRef]
- Deng, Z.D.; Lu, J.; Myjak, M.J.; Martinez, J.J.; Tian, C.; Morris, S.J.; Carlson, T.J.; Zhou, D.; Hou, H. Design and Implementation of a New Autonomous Sensor Fish to Support Advanced Hydropower Development. *Rev. Sci. Instrum.* 2014, *85*, 115001. [CrossRef] [PubMed]
- Deng, Z.; Carlson, T.; Duncan, J.P.; Richmond, M. Six-Degree-of-Freedom Sensor Fish Design and Instrumentation. Sensors 2007, 7, 3399–3415. [CrossRef] [PubMed]
- 93. Fu, T.; Deng, Z.D.; Duncan, J.P.; Zhou, D.; Carlson, T.J.; Johnson, G.E.; Hou, H. Assessing Hydraulic Conditions through Francis Turbines Using an Autonomous Sensor Device. *Renew. Energy* **2016**, *99*, 1244–1252. [CrossRef]
- 94. Martinez, J.J.; Deng, Z.D.; Titzler, P.S.; Duncan, J.P.; Lu, J.; Mueller, R.P.; Tian, C.; Trumbo, B.A.; Ahmann, M.L.; Renholds, J.F. Hydraulic and Biological Characterization of a Large Kaplan Turbine. *Renew. Energy* **2019**, *131*, 240–249. [CrossRef]
- 95. Ventikos, Y.; Sotiropoulos, F.; Patel, V.C. Modelling Complex Draft-Tube Flows Using Near-Wall Turbulence Closures. In *Hydraulic Machinery and Cavitation*; Springer: Dordrecht, The Netherlands, 1996; pp. 140–149.
- 96. Keller, M.; Sick, M.; Grunder, R.; Grafenberger, P. CFD Based Assessment of Fish-Friendliness of the Time Dependent Flow Field in a Kaplan Runner. In *HydroVision 2006*; HCI Publications: Kansas City, MO, USA, 2006.

- 97. Brown, R.S.; Ahmann, M.L.; Trumbo, B.A.; Foust, J. Fish Protection: Cooperative Research Advances Fish-Friendly Turbine Design; Pacific Northwest National Lab. (PNNL): Richland, WA, USA, 2012.
- Richmond, M.C.; Serkowski, J.A.; Ebner, L.L.; Sick, M.; Brown, R.S.; Carlson, T.J. Quantifying Barotrauma Risk to Juvenile Fish during Hydro-Turbine Passage. *Fish. Res.* 2014, 154, 152–164. [CrossRef]
- Romero-Gomez, P.; Salalila, A.; Deng, D.Z.; Peyreder, R. Evaluation of Fish-Related Properties of Kaplan Turbines at the Design Phase: Simulation-Based Outcomes vs. Experimental Data. *IOP Conf. Ser. Earth Environ. Sci.* 2022, 1079, 012016. [CrossRef]
- 100. Romero-Gomez, P.; Lang, M.; Michelcic, J.; Weissenberger, S. Particle-Based Evaluations of Fish-Friendliness in Kaplan Turbine Operations. *IOP Conf. Ser. Earth Environ. Sci.* **2019**, 240, 042016. [CrossRef]
- 101. Richmond, M.C.; Romero-Gomez, P. Fish Passage through Hydropower Turbines: Simulating Blade Strike Using the Discrete Element Method. *IOP Conf. Ser. Earth Environ. Sci.* 2014, 22, 62010. [CrossRef]
- Pan, Q.; Zhang, D.; Shi, W.; van Esch, B.P.M. Fish Damage Assessment during the Passage through Traditional and Fish-Friendly Axial-Flow Pumps with Lagrangian Tracking Approach. *Ocean Eng.* 2022, 253, 111188. [CrossRef]
- 103. Benigni, H.; Schneider, J.; Reckendorfer, W.; Jaberg, H.; Zenz, G.; Tuhtan, J. Numerical Simulation and Experimental Verification of Downstream Fish Migration in a Kaplan Turbine. *IOP Conf. Ser. Earth Environ. Sci.* 2021, 774, 012149. [CrossRef]
- Sharma, A.; Ananthan, S.; Sitaraman, J.; Thomas, S.; Sprague, M.A. Overset Meshes for Incompressible Flows: On Preserving Accuracy of Underlying Discretizations. J. Comput. Phys. 2021, 428, 109987. [CrossRef]
- 105. Koukouvinis, P.K.; Anagnostopoulos, J. A Fast 6-DoF Tracking Method for Submerged Bodies: Application to Fish Passage through a Turbine. In Proceedings of the 20th International Conference of Numerical Analysis and Applied Mathematics, Heraklion, Greece, 19–25 September 2022.
- 106. Kim, W.; Choi, H. Immersed Boundary Methods for Fluid-Structure Interaction: A Review. Int. J. Heat Fluid Flow 2019, 75, 301–309. [CrossRef]
- Huang, Z.; Cheng, Y.; Wu, J.; Diao, W.; Huai, W. FSI Simulation of Dynamics of Fish Passing through a Tubular Turbine Based on the Immersed Boundary-Lattice Boltzmann Coupling Scheme. J. Hydrodyn. 2022, 34, 135–147. [CrossRef]
- 108. Koukouvinis, P.F.; Anagnostopoulos, J. Simulating Fish Motion through a Diagonal Reversible Turbine. Energies 2023, 16, 810. [CrossRef]
- Kassanos, I.; Alexopoulos, V.; Anagnostopoulos, J. Numerical Design Methodology for Reversible Deriaz Turbine with High Energy Performance and Reduced Fish Impacts. *IOP Conf. Ser. Earth Environ. Sci.* 2022, 1079, 012076. [CrossRef]
- Ferguson, J.W.; Ploskey, G.R.; Leonardsson, K.; Zabel, R.W.; Lundqvist, H. Combining Turbine Blade-Strike and Life Cycle Models to Assess Mitigation Strategies for Fish Passing Dams. *Can. J. Fish. Aquat. Sci.* 2008, 65, 1568–1585. [CrossRef]
- Hecker, G.E.; Allen, G.S. Approach to Predicting Fish Survival for Advanced Technology Turbines. In *Hydro Review*; HCI Publications, Inc.: St. Louis, MO, USA, 2005.
- 112. Cada, G.F. The Development of Advanced Hydroelectric Turbines to Improve Fish Passage Survival. *Fisheries* **2001**, 26, 14–23. [CrossRef]
- 113. Hogan, T.W.; Cada, G.F.; Amaral, S.V. The Status of Environmentally Enhanced Hydropower Turbines. Fisheries 2014, 39, 164–172. [CrossRef]
- 114. Normandeau Associates; Skalski, J.R.; Mid-Columbia Consulting, I. Direct Passage Survival and Condition of Juvenile Chinook Salmon Passed through an Existing and New Minimum Gap Runner Turbines at Bonneville Dam, First Powerhouse, Columbia River; U.S. Army Corps of Engineers: Portland, OR, USA, 2000.
- 115. Quaranta, E.; Bahreini, A.; Riasi, A.; Revelli, R. The Very Low Head Turbine for Hydropower Generation in Existing Hydraulic Infrastructures: State of the Art and Future Challenges. *Sustain. Energy Technol. Assess.* **2022**, *51*, 101924. [CrossRef]
- Airody, A.; De Montmorency, D.; Peterson, S.D. Design Optimization of a Vaneless "Fish-Friendly" Swirl Injector for Small Water Turbines. J. Fluids Eng. 2017, 139, 091105. [CrossRef]
- 117. Cook, T.; Hecker, G.E.; Amaral, S.V.; Stacy, P.S.; Lin, F.; Taft, E.P. *Final Report—Pilot Scale Tests Alden/Concepts NREC Turbine U.S.*; Prepared for US Department of Energy: Holden, MA, USA, 2003; Volume 224-03/E63.
- Erinofiardi, E.; Koirala, R.; Shiwakoti, N.; Date, A. Sustainable Power Generation Using Archimedean Screw Turbine: Influence of Blade Number on Flow and Performance. *Sustainability* 2022, 14, 15948. [CrossRef]
- YoosefDoost, A.; Lubitz, W. Archimedes Screw Turbines: A Sustainable Development Solution for Green and Renewable Energy Generation—A Review of Potential and Design Procedures. *Sustainability* 2020, 12, 7352. [CrossRef]
- Havn, T.; Sæther, S.; Thorstad, E.; Teichert, M.A.K.; Heermann, L.; Diserud, O.H.; Borcherding, J.; Tambets, M.; Økland, F. Downstream Migration of Atlantic Salmon Smolts Past a Low Head Hydropower Station Equippped with Archimedes Screw and Francis Turbines. *Ecol. Eng.* 2017, 105, 262–275. [CrossRef]
- 121. Pauwels, I.S.; Baeyens, R.; Toming, G.; Schneider, M.; Buysse, D.; Coeck, J.; Tuhtan, J.A. Multi-Species Assessment of Injury, Mortality, and Physical Conditions during Downstream Passage through a Large Archimedes Hydrodynamic Screw (Albert Canal, Belgium). Sustainability 2020, 12, 8722. [CrossRef]
- 122. Piper, A.T.; Rosewarne, P.J.; Wright, R.M.; Kemp, P.S. The Impact of an Archimedes Screw Hydropower Turbine on Fish Migration in a Lowland River. *Ecol. Eng.* **2018**, *118*, 31–42. [CrossRef]
- Müller, S.; Cleynen, O.; Hoerner, S.; Lichtenberg, N.; Thévenin, D. Numerical Analysis of the Compromise between Power Output and Fish-Friendliness in a Vortex Power Plant. J. Ecohydraulics 2018, 3, 86–98. [CrossRef]

- 124. Haryadi; Subarjah, A.M. Sugianto Experimental Study on 3D Vortex Gravitational Turbine Runner; AIP Publishing LLC: Melville, NY, USA, 2020; p. 020025.
- 125. Brown, E.; Sulaeman, S.; Quispe-Abad, R.; Müller, N.; Moran, E. Safe Passage for Fish: The Case for in-Stream Turbines. *Renew. Sustain. Energy Rev.* **2023**, *173*, 113034. [CrossRef]
- 126. Torresi, M.; Fortunato, B.; Camporeale, S.M. Numerical Investigation of a Darrieus Rotor for Low-Head Hydropower Generation. *Procedia Comput. Sci.* **2013**, *19*, 728–735. [CrossRef]
- 127. Jacobson, P.; Amaral, S.; Castro-Santos, T.; Giza, D.; Haro, A.; Hecker, G.; McMahon, B.; Perkins, N.; Pioppi, N. *Environmental Effects of Hydrokinetic Turbines on Fish: Desktop and Laboratory Flume Studies*; Electric Power Research Institute: Washington, DC, USA, 2012.
- 128. Quaranta, E.; Pérez-Díaz, J.I.; Romero–Gomez, P.; Pistocchi, A. Environmentally Enhanced Turbines for Hydropower Plants: Current Technology and Future Perspective. *Front. Energy Res.* **2021**, *9*, 1–6. [CrossRef]
- 129. March, P.; Jacobson, P. Industry Experience with Aerating Turbines. In Proceedings of the HydroVision, Moscow, Russia, 3–5 March 2015; p. 33.

Disclaimer/Publisher's Note: The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.